Author’s response to reviews

Title: Association between Migration and Cognitive Status among Middle-Aged and Older Adults: A systematic review

Authors:

Hanzhang Xu (hx20@duke.edu)
Yinan Zhang (yinan.y.zhang@duke.edu)
Bei Wu (bei.wu@nyu.edu)

Version: 3 Date: 08 Aug 2017

Author’s response to reviews:

We would like to thank the editor and the reviewers for their careful consideration of our manuscript. We have revised the manuscript according to their suggestions, and we believe these revisions have substantially improved the quality of our work. Below, we describe point-by-point how we addressed each comment. We highlighted the changes with red text in the revised manuscript.

1. Please note that as data was used in this manuscript, you cannot state “Not Applicable” in the Availability of data and materials section. Please amend this statement to “All data generated or analysed during this study are included in this published article”.

RESPONSE: We've made changes accordingly.

2. Please provide figure titles/legends under a separate heading of 'Figure Legends' after the References. If Figure titles/legends are within the main text of the manuscript, please move them.

Figure files should contain only the image/graphic, as well as any associated keys/annotations. If titles/legends are present within the figure files, please remove them.

Figures should be provided as separate files, and each figure of a manuscript should be submitted as a single file.

RESPONSE: We've made changes accordingly.
3. Please change the Findings heading to Results in the Abstract.

RESPONSE: We've made changes accordingly.

4. Please change the Introduction heading to Background in the main text.

RESPONSE: We've made changes accordingly.

5. At this stage, please upload your manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours. All relevant tables/figures/additional files should also be clean versions. Figures (and additional files) should remain uploaded as separate files.

RESPONSE: We've made changes accordingly. All the files are clean versions.